Title: On line clinical reasoning assessment with the script concordance test: results of a french pilot study

Authors:

Louis Sibert (louis.sibert@chu-rouen.fr)
Stefan J. Darmoni (stefan.darmoni@chu-rouen.fr)
Badisse Dahamna (badisse.dahamna@chu-rouen.fr)
Marie-France Hellot (Marie-France.Hellot@chu-rouen.fr)
Jacques Weber (jacques.weber@chu-rouen.fr)
Bernard Charlin (bernard.charlin@umontreal.ca)

Version: 3 Date: 25 August 2006

Author's response to reviews: see over
MS: 1818731618105034
On line clinical reasoning assessment with Script Concordance test in Urology: Results of a French pilot study.

Dear Sir

Thank you for your e-mail dated August 24. In reply, please find enclosed a point-by-point description of the changes made, as well as a revised version of the manuscript.

1) All the Web site addresses have now been removed from the manuscript text. They were cited in the reference list as mentioned in the guidelines.

2) A "Conclusion" section has now been added in the manuscript text.

3) As suggested by the Editorial Production Team, the figure 1 has now been deleted from the manuscript.

Thank you for considering our article for publication in BMC Medical Education.

With best regards,

Louis Sibert, MD, Ph.D.